Author's response to reviews

Title: Repetition of suicide attempts across episodes of severe depression
Behavioural sensitisation found in suicide group but not in controls

Authors:

Louise Brådvik (louise@bradvik.se)
Mats Berglund (Mats.Berglund@med.lu.se)

Version: 8 Date: 5 January 2011

Author's response to reviews:

Point-by-point typo errors in manuscript
Repetition of suicide attempts across episodes of severe depression
Behavioural sensitisation found in suicide group but not in controls
MS ID: 2032317877465071

P 2 (abstract) Conclusion moved down to be a new §
P 3 line 15 these differences were changed to this difference was (there was only one difference in the previous sentence as opposed to three similarities)
P3 line 18 Suicides is substituted with People who died by suicide… in agreement with the previous suggestion of the reviewer who wanted to avoid the use of the term “suicide” for human beings.
P7 second § (course of depression) 2nd line two commas are added.
References: 20 title is now bolded.

All changes are marked in red.